Author's response to reviews

Title: Predictors of positive response to cardiac resynchronization therapy

Authors:

Diana Rinkuniene (diana.rinkuniene@gmail.com)
Silvija Bucyte (sansilveria@gmail.com)
Kristina Ceseviciute (merginakristina@gmail.com)
Silvijus Abramavicius (silvijus2@yahoo.com)
Kristina Baronaite-Dudoniene (zuviene@yahoo.com)
Jolanta Laukaitiene (jolanta_laukaitiene@yahoo.co.uk)
Tomas Kazakevicius (kazakev@gmail.com)
Vytautas Zabiela (zabiela.vytautas@kmuk.lt)
Vytautas Sileikis (sileikis@one.lt)
Aras Puodziukynas (aras.puodziukynas@gmail.com)
Renaldas Jurkevicius (Renaldas.Jurkevicius@lsmuni.lt)

Version: 2
Date: 15 April 2014

Author's response to reviews:

Dear Madam,

Thank you for the opportunity to improve the quality of our manuscript. We thank the reviewers for their efforts. This documents contains a point-by-point answer to all the questions. In addition to that we have submitted a new version of the manuscript. Changes are depicted in red. We hope the answers will be satisfactory.

Sincerely yours,
Diana Rinkuniene